Author's response to reviews

Title: Clinicians' evaluations of, endorsements of, and intentions to use practice guidelines change over time: a retrospective analysis from an organized guideline program

Authors:

Melissa Brouwers (mbrouwer@mcmaster.ca)
Steven Hanna (hannas@mcmaster.ca)
Mona Abdel-Motagally (abdelmm@mcmaster.ca)
Jennifer Yee (jennifer.yee@sunnybrook.ca)

Version: 2 Date: 4 September 2008

Author's response to reviews:

September 4, 2008

Drs. Martin Eccles and Brian Mittman
Co-Editors, Implementation Science

Dear Dr. Eccles and Mittman:

I have made a few minor grammatical and editorial changes as follows:

Page 2 - Abstract - Line 2 - grammatical change
Page 7 - Results, 1st paragraph - editorial change (guideline to CPG)
Page 8 - Quality, endorsement, etc. - 2nd Paragraph, last line - grammatical change
Page 10 - Belief CPGs - Last sentence - editorial change
Page 10 - Misconception beliefs - Paragraph 1, 2nd sentence - editorial change
Page 14 - Paragraph 2, 5th sentence - editorial change

If you have any questions, please contact me. Thank you.

Dr. Melissa Brouwers